Author's response to reviews

Title: The prevalence and correlates of alcohol use and alcohol use disorders: a population based study in Sri Lanka

Authors:

Helena M.S. Zavos (helena.zavos@kcl.ac.uk)
Sisira Siribaddana (nipuna@stmail.lk)
Harriet A Ball (harriet.ball@gmail.com)
Micheal T Lynskey (micheal.lynskey@kcl.ac.uk)
Athula Sumathipala (athula.sumathipala@kcl.ac.uk)
Fruhling V Rijsdijk (fruhling.rijsdijk@kcl.ac.uk)
Matthew Hotopf (matthew.hotopf@kcl.ac.uk)

Version: 3 Date: 27 November 2014

Author's response to reviews:

We have confirmed the consent procedure. Please see edit below:

The consent procedure was the same for all participants including those under 18 years old and was approved by the relevant ethics committees. Participants were included only if they provided written informed consent.